We added the "Availability of supporting data" section and reference [22] as required on the letter dated Dec 13.

In addition, we checked the mockup of the DOI. Everything looks great. Please go ahead to switch it live.

Thank you for your consideration. Ka Yee Yeung Associate Professor, University of Washington Tacoma